Author's response to reviews

Title: Epidemiology of smoking during pregnancy among Canadian women: A national survey

Authors:

Ban Al-Sahab (bsahab@yorku.ca)
Masarat Saqib (masarat@yorku.ca)
Gabriel Hauser (ghauser@yorku.ca)
Hala Tamim (htamim@yorku.ca)

Version: 2 Date: 14 August 2009

Author's response to reviews:

The manuscript has been revised to include information on the ethics approval and consent form.